\def\papertitle{Results in Semi-$E$-Convex Functions}
\def\paperauthorA{Ayache Benhadid}
%\def\paperauthorB{Author_B_fullname}
% you can add more authors using alphabets after \paperauthor
% 
%------------------------------------------------------------------------------------------
% Do not delete this part starting from here and up to .....
\documentclass[oneside]{article}
\usepackage{icma2021,amssymb,amsmath,mathrsfs} 
%\setcounter{page}{1}
%\input{../../expages.tex}\setpagenumber{1}
%\input{expages.tex}\setpagenumber{1}
\ninept

\usepackage{mathptmx} 
% Saves a lot of ouptut space in PDF... after conversion with the distiller	  
% Delete if you cannot get PS fonts working on your system.

% pdf-tex settings: detect automatically if run by latex or pdflatex
\newif\ifpdf
\ifx\pdfoutput\relax
\else
\ifcase\pdfoutput
\pdffalse
\else
\pdftrue
\fi

\ifpdf % compiling with pdflatex
\usepackage[pdftex,
pdftitle={\papertitle},
%pdfauthor={\paperauthorA},
% for more than one author use pdfauthor={\paperauthorA,\paperauthorB, .....},
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%ù
colorlinks=false, % links are activated as colror boxes instead of color text
bookmarksnumbered, %use section numbers with bookmarks
pdfstartview= XYZ	% starts with zoom=100% instead of full screen; especially useful if working with a big screen :-)
]{hyperref}
\pdfcompresslevel=9
\usepackage[pdftex]{graphicx}
\usepackage[figure,table]{hypcap}
\else % compiling with latex
\usepackage[dvips]{graphicx}
\usepackage[dvips,
colorlinks=false,           % no color links
bookmarksnumbered, %use section numbers with bookmarks
pdfstartview= XYZ	% starts with zoom=100% instead of full screen
]{hyperref}
% hyperrefs are active in the pdf file after conversion
\usepackage[figure,table]{hypcap}
\fi

\newcommand\dsum{\displaystyle\sum}
\newcommand\dprod{\displaystyle\prod}
%------------------------------------------------------------------------------------------
\title{\papertitle}
\affiliation{\paperauthorA}	% This command replaces \name{The DAFx Crew} 
{Abbes Laghrour University, Faculty of Science and Technology\\ Departement of Mathematics and Informatics, Khenchela, Algeria.\\
	 e-mail: benhadidayache@yahoo.fr}

%\usepackage{layout}
%------------------------------------------------------------------------------------------
%%%%%%%%%%%%%%%%%%%%%%% to here %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document}
\begin{thebibliography}{}
\bibitem{Ref2} X. Chen, Some properties of semi-E-convex functions, Journal of mathematical Anaysis andApplication, 275, 251-262, (2002).
\bibitem{Ref3} E. A. Youness, E-convex sets, E-convex functions, and E-convex programming, Journal of Optimization Theory and Applications, 102, 439-450, (1999).
\bibitem{Ref4}G. Dal maso, An Introduction to ${\Gamma}$-convergence,10-11. BirKhauser, Boston, (1993).
\bibitem{Ref1} A. A. Megahed, H. G. Gomma, E. A. Youness, A. H. Banna, Optimality conditions of $E$-convex programing for an $E$-differentiable function, journal of Inequalities and Applications, 246, 1-11, (2013).
\bibitem{Ref5} M. A. Noor, M. U. Awan, K. I. Noor, On some inequalities for relative semi-convex functions, journal of Inequalities and Applications, 332, 1-16,(2013).
\end{thebibliography}

	
	%\layout
	
	% more pdf-tex settings:
	\ifpdf % used graphic file format for pdflatex
	\DeclareGraphicsExtensions{.png,.jpg,.pdf}
	\else  % used graphic file format for latex
	\DeclareGraphicsExtensions{.eps}
	\fi
	
	\maketitle
	
\begin{abstract}
The concept of convexity and its various generalizations is important for quantitative and qualitative studies in operations research or applied mathematics. Recently, E-convex sets and E-convex functions were introduced by Youness \cite{Ref3}, and they have some important applications in various branches of mathematical sciences. Youness in \cite{Ref3} introduced a class of sets and functions which is called $E$-convex sets and $E$-convex functions by relaxing the definition of convex sets and convex functions. Xiusu Chen \cite{Ref2} introduced a new concept of semi $E$-convex functions and discussed its properties. According to Xiusu Chen \cite{Ref2}, if a function $f:M\rightarrow \mathbb{R}$ is semi-$E$-convex on an $E$-convex set $M\subset \mathbb{R}^{n}$ then, $f(E(x))\leq f(x)$ for each $x\in M$. In this article we have discussed the inverse of this proposition and present some results for convex functions.
\end{abstract}
\section{Introduction}
\indent
Youness in \cite{Ref3} introduced a class of sets and functions which is called
$E$-convex sets and $E$-convex functions by relaxing the definition of convex
sets and convex functions. Following this Xiusu Chen \cite{Ref2} introduced a new
class of semi-$E$-convex functions and applied these functions to non linear
programming problems see for instance \cite{Ref1,Ref5} . In this paper, we give weak condition for a lower
semi-continuous function on $\mathbb{R}^{n}$ to be a semi-$E$-convex function, we also present some
results for convex functions.
\section{Preliminaries}

Let $M$ be a nonempty subset of $%
\mathbb{R}^{n}$ and let $E$ : $\mathbb{R}^{n}\rightarrow \mathbb{R}^{n}$ be
a map. We recall:
\begin{definition}  \cite{Ref3} A set $M\subseteq
\mathbb{R}^{n}$ is said to be $E$-convex in $\mathbb{R}^{n}$ if
\[
tE(x)+(1-t)E(y)\in M,
\]
for each $x,y\in M$ and all $t\in \left[ 0,1\right] $.
\end{definition}
\begin{definition}\cite{Ref3} A function $f:M\rightarrow \mathbb{R}$ is said to be $E$-convex on $M$ if $M$
is $E$-convex and
%\begin{equation*}
\[
f\left( tE(x)+(1-t)E(y)\right) \leq tf\left( E\left( x\right) \right)
+(1-t)f\left( E\left( y\right) \right) ,
\]
%\end{equation*}
for each $x,y\in M$ and all $t\in \left[ 0,1\right] $.
\end{definition}
\begin{definition}\cite{Ref2}\label{ds}  A function $f:M\rightarrow \mathbb{R}$ is said to be semi-$E$-convex on $M$ if
$M$ is $E$-convex and
\[
f\left( tE(x)+(1-t)E(y)\right) \leq tf\left( x\right) +(1-t)f\left( y\right),
\]
for each $x,y\in M$ and all $t\in \left[ 0,1\right] $.
\end{definition}
\begin{definition}\cite{Ref2} We define a map $E\times I$ as follows:
\[ \left.
\begin{array}{rcl}
E\times I:\mathbb{R}^{n}\times \mathbb{R} & \rightarrow & \mathbb{R}^{n}\times \mathbb{R} \\
\left(x,t\right) & \rightarrow & (E\times I)(x,t)=\left( E(x),t\right).
\end{array}%
\right.\]
\end{definition}
This Proposition gives a characterization of a semi-$E$-convex function in
term of its $epi\left( f\right) $.
\begin{proposition}\label{p1} \cite{Ref2}   The function $ f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ is semi-$E$-convex on $%
\mathbb{R}^{n}$ if and only if  its epigraph $epi\left( f\right) =\left\{ \left( x,\alpha
\right) \in \mathbb{R}^{n}\times \mathbb{R}:f\left( x\right) \leq \alpha
\right\} $ is $E\times I$-convex
on $\mathbb{R}^{n}\mathbb{\times R}$.
\end{proposition}

\begin{definition} \label{p2} \cite{Ref4} A function $f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ is lower semi-continuous if and only if, for every real number $%
\alpha $, the set $\left\{ x\in\mathbb{R}^{n}:f\left( x\right) \leq \alpha \right\} $ is closed.
\end{definition}
In the following we introduce a Proposition about lower semi-continuous
functions which shall be used in the sequal. For details and for the missing proofs we refer G. Dal Maso  \cite{Ref4}.
\begin{proposition} \label{p3} \cite{Ref4} A function $f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ is lower semi-continuous
if and only if its epigraph is closed.
\end{proposition}
\begin{definition} Let $(x,s),(y,t)\in \mathbb{R}^{n+1}$, with $x,y\in \mathbb{R}^{n}$ and $%
s,t\in\mathbb{R}$. The line segment $[(x,s),(y,t)]$ (with endpoints $(x,s)$ and $%
(y,t))$ is the segment
\[
\left\{ \alpha (x,s)+(1-\alpha )(y,t):0\leq \alpha \leq 1\right\} .
\]
If $(x,s)\neq (y,t)$, the interior $](x,s),(y,t)[$ of $[(x,s),(y,t)]$ is the
segment
\[
\left\{ \alpha (x,s)+(1-\alpha )(y,t):0<\alpha <1\right\} .
\]
In a similar way, we can define $[(x,s),(y,t))$ and $((x,s),(y,t)]$.

\end{definition}%
\section{Main results for semi-$E$-convex functions}
\begin{lemma}\label{l1}
Let $E:\mathbb{R}^{n}\rightarrow \mathbb{R}^{n}$ be a linear and idempotent
map. Consider \\$\left( \overline{x},u\right) \in \lbrack (E\left( x\right)
,s),(E\left( y\right) ,t)].$ Then%
\begin{equation*}
E\left( \overline{x}\right) =\overline{x}.
\end{equation*}
\end{lemma}
\begin{proof} Let $\left( \overline{x},u\right) \in \lbrack (E\left( x\right)
,s),(E\left( y\right) ,t)], $ then there exist $\alpha\in \left[ 0,1\right] $, such that \\$\left( \overline{x},u\right)=\alpha (E\left( x\right) ,s)+(1-\alpha )(E\left( y\right) ,t). $
 % \end{eqnarray* }
Using the fact that $E$ is linear and idempotent map, we have
\begin{eqnarray*}
(E\times I)\left( \overline{x},u\right) &=&\left( E\left( \alpha E\left(
x\right) +(1-\alpha )E\left( y\right) \right) ,\alpha s+(1-\alpha )t\right)
\\
&=&\left( \alpha E\left( x\right) +(1-\alpha )E\left( y\right) ,\alpha
s+(1-\alpha )t\right) \\
&=&\left( \overline{x},u\right).
\end{eqnarray*}
On the other hand
$(E\times I)\left( \overline{x},u\right) =\left( E\left( \overline{x}\right), u\right)$,
therfore $E\left( \overline{x}\right) =\overline{x}$.
\end{proof}

We shall make use the following three sets:\newline
\begin{align}
H_{Sci}&=\left\{ f:\mathbb{R}^{n}\rightarrow \mathbb{R},\mbox{ $f$ is lower
semi continuous}\right\}, \\
H_{L,I}&=\left\{ E:\mathbb{R}^{n}\rightarrow \mathbb{R}^{n},\,E \mbox{
is linear and idempotent}\right\} \\
\intertext{and for each $E\in H_{L,I}$ we define $H_{E}$ as follows:}
H_{E}&=\left\{ f\in H_{Sci},\text{ }f\left( E\left( x\right) \right)\leq
f\left( x\right) \mbox{ for all }x\in \mathbb{R}^{n}\right\}
\end{align}
\begin{theorem}\label{t1}
Let $E\in H_{L,I}$, and $f\in H_{E}$. Suppose that there
exists an $\alpha \in \left] 0,1\right[ $ such that for all $x,y\in \mathbb{R%
}^{n}$, $s,t\in \mathbb{R}$ such that $f\left( x\right) <s,$ $f\left(
y\right) <t,$ 
\[
f\left( \alpha E(x)+(1-\alpha )E(y)\right) <\alpha s+(1-\alpha )t.
\]
Then $f$ is semi-$E$-convex.
\end{theorem}
\begin{proof}  By Proposition (\ref{p1}), it is sufficient to show that $epi\left( f\right) $ is $%
E\times I$-convex as a subset of $\mathbb{R}^{n}\mathbb{\times R}$. By
contradiction, suppose that there exist $(x_{1},\alpha _{1}),(x_{2},\alpha
_{2})\in epi\left( f\right) $ (with $x_{1},x_{2}\in \mathbb{R}^{n}$ and $%
\alpha _{1},\alpha _{2}\in \mathbb{R}$) and $\alpha _{0}\in \left] 0,1\right[
$ such that, \\$\left( \alpha _{0}E\left( x_{1}\right) +\left( 1-\alpha
_{0}\right) E\left( x_{2}\right) ,\alpha _{0}\alpha _{1}+\left( 1-\alpha
_{0}\right) \alpha _{2}\right) \notin epi\left( f\right) .\\$ Let $%
x_{0}=\alpha _{0}E\left( x_{1}\right) +\left( 1-\alpha _{0}\right) E\left(
x_{2}\right) $ and $\lambda _{0}=\alpha _{0}\alpha _{1}+\left( 1-\alpha
_{0}\right) \alpha _{2}$, then $\left( x_{0},\lambda _{0}\right) \notin
epi\left( f\right) $. Using the fact that $f\in H_{ E }$, we
see that $(E\left( x_{1}\right) ,\alpha _{1}),(E\left( x_{2}\right) ,\alpha
_{2})\in epi\left( f\right) . $ Let%
%\begin{equation*}
\[
A=epi\left( f\right) \cap \left[ (E\left( x_{1}\right) ,\alpha _{1}),\left(
x_{0},\lambda _{0}\right) \right]
\]
%\end{equation*} %
 and %
%\begin{equation*}
\[
B=epi\left( f\right) \cap \left[ \left( x_{0},\lambda _{0}\right) ,(E\left(
x_{2}\right) ,\alpha _{2})\right].
\]
%\end{equation*}  %
Since $f\in H_{Sci}$, by Proposition (\ref{p3}), $epi\left( f\right) $ is a closed
subset of $\mathbb{R}^{n}\mathbb{\times R}$. Consequently, $A$ and $B$ are
bounded and closed subsets of $\mathbb{R}^{n}\mathbb{\times R}$.\\ Also we have $\left(
x_{0},\lambda _{0}\right) \notin A$ and $\left( x_{0},\lambda _{0}\right)
\notin B$. Thus there exist $Z_{A}=\left( x_{3},\alpha _{3}\right) \in A$
and $Z_{B}=\left( x_{4},\alpha _{4}\right) \in B$ such that,
\[\min\limits_{Z\in A}\left\Vert Z-\left( x_{0},\lambda _{0}\right)\right\Vert =\left\Vert Z_{A}-\left( x_{0},\lambda _{0}\right) \right\Vert \]
and \[\min\limits_{Z\in B}\left\Vert Z-\left( x_{0},\lambda _{0}\right)\right\Vert =\left\Vert Z_{B}-\left( x_{0},\lambda _{0}\right) \right\Vert. \]
Hence, we have
\begin{equation}
\left] Z_{A},Z_{B}\right[ \cap epi\left( f\right) =\emptyset. \label{ZaZb1}
\end{equation}
On the other hand, since $Z_{A}\in epi\left( f\right) $ and $Z_{B}\in
epi\left( f\right) $, we get\newline
$f\left( x_{3}\right) <\alpha _{3}+\varepsilon ,$ $f\left( x_{4}\right)
<\alpha _{4}+\varepsilon $ for each $\varepsilon >0.$\newline
Since $\alpha \left( \alpha _{3}+\varepsilon \right) +(1-\alpha )\left(
\alpha _{4}+\varepsilon \right) =\alpha \alpha _{3}+(1-\alpha )\alpha
_{4}+\varepsilon.$ By the hypothesis of the theorem, we obtian
%\begin{equation*}
\[
f\left( \alpha E(x_{3})+(1-\alpha )E(x_{4})\right) <\alpha \alpha
_{3}+(1-\alpha )\alpha _{4}+\varepsilon.
\]
%\end{equation*}%
Since $\varepsilon $ is an arbitrary positive real number, it follows that%
\begin{equation}
f\left( \alpha E(x_{3})+(1-\alpha )E(x_{4})\right) \leq \alpha \alpha
_{3}+(1-\alpha )\alpha _{4}.\label{q}
\end{equation}%
Since $Z_{A}\in A\subset \left[ \left( E\left( x_{1}\right) ,\alpha
_{1}\right) ,(E\left( x_{2}\right) ,\alpha _{2})\right] $ and \newline
$Z_{B}\in B\subset \left[ \left( E\left( x_{1}\right) ,\alpha _{1}\right)
,(E\left( x_{2}\right) ,\alpha _{2})\right] $. By Lemma  (\ref{l1}) we have $E\left(
x_{3}\right) =x_{3}$ and \\$E\left( x_{4}\right) =x_{4}.$ Using (\ref{q}) we get%
\[
\left( \alpha x_{3}+(1-\alpha )x_{4},\alpha \alpha _{3}+(1-\alpha )\right)
\alpha _{4})\in epi\left( f\right).
\] %
Therfore%
\[
\alpha Z_{A}+(1-\alpha )Z_{B}\in epi\left( f\right)
\] %
which contradicts (\ref{ZaZb1}). Thus, we conclude that $epi\left( f\right) $ is $%
E\times I$-convex.
\end{proof}
\begin{theorem}\label{t5}
Let $E:\mathbb{R}^{n}\rightarrow \mathbb{R}^{n}$ be a linear and idempotent
map, $f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ be lower semi-continuous and $%
f\left( E\left( x\right) \right) \leq f\left( x\right)$ for all $x\in 
\mathbb{R}^{n}$. Then $f$ is semi-$E$-convex if and only if there exists an $%
\alpha \in \left] 0,1\right[ $ such that for all $x,y\in \mathbb{R}^{n}$
\[
f\left( \alpha E(x)+(1-\alpha )E(y)\right) \leq \alpha f\left( x\right)
+(1-\alpha )f\left( y\right).
\]
\end{theorem}
\begin{proof} Follows from Theorem (\ref{t1}) with $s=f\left( x\right) +\varepsilon $ and $t=f\left( y\right) +\varepsilon $ for each $\varepsilon >0,$ then taking $\varepsilon \rightarrow 0.$
\end{proof}
\begin{corollary}
Let $E:\mathbb{R}^{n}\rightarrow \mathbb{R}^{n}$ be a linear and idempotent
map, $f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ be lower semi-continuous and $%
f\left( E\left( x\right) \right) \leq f\left( x\right) $ for all $x\in 
\mathbb{R}^{n}$. Then $f$ is semi-$E$-convex if and only if for all $x,y\in
\mathbb{R}^{n}$,%
\[
f\left( \frac{1}{2}\left( E(x)+E(y)\right) \right) \leq \frac{1}{2}\left[
f\left( x\right) +f\left( y\right) \right] .
\]
\end{corollary}

\begin{theorem}\label{t6}
Let $E:\mathbb{R}^{n}\rightarrow \mathbb{R}^{n}$ be a linear and idempotent
map, $f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ be lower semi-continuous and $%
f\left( E\left( x\right) \right) \leq f\left( x\right) $ for all $x\in 
\mathbb{R}^{n}$. Then f is semi-$E$-convex
if and only if for all $x,y\in \mathbb{R}^{n}$, there exists an $\alpha \in %
\left] 0,1\right[ $ ($\alpha $ depends on $x,y$) such that%
\begin{equation}
f\left( \alpha E(x)+(1-\alpha )E(y)\right) \leq \alpha f\left( x\right)\label{ZaZb3}
+(1-\alpha )f\left( y\right).
\end{equation}
\end{theorem}


If we take $E=Id_{\mathbb{R}^{n},}$ we get $E\in H_{L,I}$, and $H_{E}=H_{Sci}.$ Then we find results about convex functions.

\begin{theorem}  Let  $f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ be lower
semi-continuous.Then f is convex if and only if there exists an $%
\alpha \in \left] 0,1\right[ $ such that, for all $x,y\in \mathbb{R}^{n}$,%
\[
f\left( \alpha x+(1-\alpha )y\right) \leq \alpha f\left( x\right) +(1-\alpha
)f\left( y\right) .
\]%
\end{theorem}
\begin{theorem} Let  $f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ be lower
semi-continuous. Then f is convex if and only if for all $x,y\in
\mathbb{R}^{n}$, there exists an $\alpha \in \left] 0,1\right[ $ ($\alpha $
depends on $x,y$) such that%
\[
f\left( \alpha x+(1-\alpha )y\right) \leq \alpha f\left( x\right) +(1-\alpha
)f\left( y\right).
\]
\end{theorem}
\begin{corollary}
 Let  $f:\mathbb{R}^{n}\rightarrow \mathbb{R}$ be lower
semi-continuous. Then $f$ is convex if and only if for all $x,y\in
\mathbb{R}^{n}$,%
\[
f\left( \frac{1}{2}\left( x+y\right) \right) \leq \frac{1}{2}\left[
f\left( x\right) +f\left( y\right) \right] .
\]
\end{corollary}


\end{document}
